Author’s response to reviews

Title: Knowledge gaps in patients with COPD and their proxies

Authors:

Nienke Nakken (nienkenakken@ciro-horn.nl)
Daisy Janssen (daisyjanssen@ciro-horn.nl)
Esther van den Bogaart (esthervdbogaart@hotmail.com)
Jean Muris (jean.muris@maastrichtuniversity.nl)
Jan Vercoulen (jan.vercoulen@radboudumc.nl)
Frank Custers (f.custers@zuyderland.nl)
Gerben Bootsma (g.bootsma@zuyderland.nl)
Michiel Gronenschild (m.gronenschild@zuyderland.nl)
Emiel Wouters (e.wouters@mumc.nl)
Martijn Spruit (martijnspruit@ciro-horn.nl)

Version: 1 Date: 30 Jan 2017

Author’s response to reviews:

The editor’s comment was that table 2-5 were missing. According to the author guideline: "• Tables less than one A4 or Letter page in length can be placed in the appropriate location within the manuscript." I placed tables 2-5 in the appropriate location within the manuscript. Tables 1 and 6 are larger than one A4 in length, so I placed them at the end of the manuscript (also according to the guidelines).